Author's response to reviews

Title: A presenting with Abnormal Fingernail Beds Following Carbon Monoxide Poisoning A Case Report and Review of the Literature

Authors:

Mehrangiz Hatami (mehrangizhatami@yahoo.com)
Frederick Naftolin (Frederick.Naftolin@nyumc.org)
Massoud A Khatamee (frfbaby@msn.com)

Version: 3 Date: 5 February 2013

Author's response to reviews: see over
Dear Dr. Michael Kidd,

This is revised manuscript entitled “A presenting with Abnormal Fingernail Beds Following Carbon Monoxide Poisoning A Case Report and Review of the Literature” for publication in “The Journal of Medical Case Reports”. To the best of the authors’ knowledge, this is the first report in the literature of fingernail bed manifestations in the survivor of carbon monoxide intoxication. Signs and symptoms of carbon monoxide poisoning are vague and may be mistaken as a migraine, flu, alcohol intoxication, and heart disease, etc... This finding should be considered in diagnosis of carbon monoxide exposure.

This is an original case report that has not been published previously and is not under consideration for publication elsewhere.

Thank you in advance for your attention to this manuscript and I look forward to hearing from you very soon.

Best regards,

Mehrangiz Hatami, MD.

P.O. BOX 77

Bronx, NY, 10461

E Mail: Mehrangizhatami@yahoo.com

Phone: (646) 705 7270